Author's response to reviews

Title: A comparison of supracostal and infracostal access approaches in treating renal and upper ureteral stones using MPCNL with the aid of a patented system

Authors:

Difu Fan (fandifu@126.com)
Leming Song (xdh888@yahoo.com)
Donghua Xie (xiedh07@gmail.com)
Min Hu (gzshm2005@126.com)
Zuofeng Peng (pzf5201@163.com)
Tairong Liu (tttrong@126.com)
Chuance Du (duchuance@126.com)
Lufeng Zhu (doctorzlf@sohu.com)
Lei Yao (yaoshisan4@163.com)
Jianrong Huang (119220468@qq.com)
Zhongsheng Yang (zhongshengyang@126.com)
Shulin Guo (gshul2007@163.com)
Wen Qin (qinwen1964@126.com)
Jiuqing Zhong (ddh418@gmail.com)
Zhangqun Ye (zhangqun_ye@yahoo.com.cn)
Xiaohui Liao (ddh418@gmail.com)

Version: 5 Date: 28 September 2015

Author's response to reviews:

September 27th, 2015
Editor, BMC Urology

Dear Editor,

On behalf of all the co-authors, please find a revised manuscript entitled “A comparison of supracostal and infracostal access approaches in treating renal and upper ureteral stones using MPCNL with the aid of a patented system” submitted to your prestigious journal BMC Urology. All the authors have agreed with the contents and the data are original without publishing in any other places. There are no competing interests.

We have answered the editor’ comments as below. Also, I have reformatted the manuscript according to the journal requirements.

EDITORIAL REQUEST:
1. Include additional context in the Abstract.

Thank you for your suggestion. We added one sentence in the Background section of the Abstract.
2. Keyword

Thank you for pointing this out. We moved the keywords from the title page to the end of the Abstract.

3. Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.

Yes, we found some parts in the sections of Reference, Table, and Figure legends were not correctly formatted and did corrections.

Thank you for your consideration!

Sincerely,

Leming Song, MD
Department of Urology
The Affiliated Ganzhou People’s Hospital of Nanchang University
17 Hongqi Avenue, Ganzhou
Jiangxi 341000, China
Tel (011-86) –797-8120226
Fax (011-86) –797-8112320
Email xdh888@yahoo.com